Author's response to reviews

Title: Therapeutic Angiogenesis in Cardiovascular Disease

Authors:

Hilal A Al Sabti (alsabti@hotmail.com)

Version: 2 Date: 1 June 2007

Author's response to reviews:

Dear Chief Editor, I had revised the manuscript. I had corrected some comments as per the reviewers comments, I also added a new section for the clinical trials.

Review 1: Dr R. Chiu
1) I had corrected page four information about vasculogenesis, it use to be thought that it happen only during embberyogenesis, but the subsequent work of Dr Asahara published late 97 and after 2000 showed that this process happen during adulthood as well.
2) I had added a section about a short review of the important clinical trials that have been published and I do agree that these trials are still quite disappointing up to date, but the cell therapy trials we still do not have much of long term data about them.
3) The subtitles in page 9 was changed.

Reviewer 2: Dr C Detter:
Dr Detter had correctly pointed that this paper needs to be updated as the references were somewhat old, had updated it by adding more information and new references by means of the clinical trials and their results and also the different modalities of delivery or treatment whether it was by intracoronary delivery or intravenous delivery or through a direct implantation. I did not go into the details of the delivery as I think this might be beyond the scope of this Review.

thank you